

# THE UNIVERSITY of EDINBURGH

# Edinburgh Research Explorer

# Carvedilol is associated with improved survival in patients with liver cirrhosis and ascites

#### Citation for published version:

Sinha, R, Lockman, KA, Mallawaarachchi, N, Robertson, M, Plevris, J & Hayes, P 2017, 'Carvedilol is associated with improved survival in patients with liver cirrhosis and ascites', *Journal of Hepatology*. https://doi.org/10.1016/j.jhep.2017.02.005

#### **Digital Object Identifier (DOI):**

10.1016/j.jhep.2017.02.005

#### Link:

Link to publication record in Edinburgh Research Explorer

**Document Version:** Peer reviewed version

Published In: Journal of Hepatology

#### General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

#### Take down policy

The University of Édinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.



#### Title page

Carvedilol use is associated with improved survival in patients with liver cirrhosis and ascites.

Rohit Sinha<sup>1,2</sup>, Khalida A Lockman<sup>2</sup>, Nethmee Mallawaarachchi<sup>2</sup>, Marcus Robertson<sup>1</sup>, John N Plevris<sup>1,2</sup>, Peter C Hayes<sup>1,2</sup>

<sup>1</sup>Liver Unit, <sup>2</sup>Hepatology Laboratory,

The Royal Infirmary of Edinburgh and The University of Edinburgh,

51 Little France Crescent, Edinburgh EH16 4SA, UK

#### Corresponding author:

Dr. Rohit Sinha

Liver Unit,

The Royal Infirmary and The University of Edinburgh,

51 Little France Crescent,

Edinburgh EH16 4SA,

United Kingdom.

Email: rohit.sinha@nhs.net

Telephone: 0044 131 2421625

Fax: 0044 131 2421633

#### Keywords:

Carvedilol; Non-selective beta-blockers; Ascites; Decompensated liver disease

Electronic word count: 3,392 words

**Total Number of Figures**: 2

**Total Number of Tables:** 5

#### Financial disclosures and conflict of interest:

Authors have no financial disclosure or conflict of interest to declare

#### Authors' contribution

RS collected data, performed statistical analysis and wrote the manuscript. KAL assisted in statistical analysis, data collection, manuscript revision and presentation. NM and MR contributed to data collection. PCH conceptualised the study. All authors reviewed and approved the final manuscript.

#### Abstract

**Background and aims:** Carvedilol, a non-selective beta-blocker (NSBB) with additional antialpha 1 receptor activity, is a potent portal hypotensive agent. It has been used as prophylaxis against variceal bleed. However, its safety in decompensated liver cirrhosis with ascites is still disputed. In this study, we examined whether long-term use of carvedilol in the presence of ascites is a risk factor for mortality.

**Methods:** This was a single-centre retrospective analysis of 325 consecutive patients with liver cirrhosis and ascites presenting to our Liver Unit between 1<sup>st</sup> of January 2009 to 31<sup>st</sup> August 2012. The primary outcome was all-cause and liver-specific mortality in patients receiving or not receiving carvedilol as prophylaxis against variceal bleeding.

**Results:** The final cohort after propensity score matching comprised 264 patients. Baseline ascites severity and UK End Stage Liver Disease (UKELD) score between carvedilol (n=132) and non-carvedilol (n=132) groups were comparable. Median follow-up time was 2.3 years. Survival at the end of the follow-up was 24% and 2% for carvedilol and non-carvedilol group respectively (Log Rank p<0.0001). The long-term survival was significantly better in carvedilol than non-carvedilol group (Log Rank p<0.001). The survival difference remained significant after adjusting for age, gender, ascites severity, aetiology of cirrhosis, previous variceal bleed, spontaneous bacterial peritonitis prophylaxis, serum albumin and UKELD with hazard ratio of 0.59 [CI 0.44, 0.80] (p=0.001), suggesting 41% reduction in mortality risk. When stratified as per the severity of ascites, carvedilol therapy resulted in hazard ratio of 0.47 [0.29, 0.77] (p=0.003) in those with mild ascites. Even with moderate or severe ascites, carvedilol use was not associated with increased mortality risk.

**Conclusion:** Long-term carvedilol therapy is not harmful in decompensated cirrhosis with ascites.

Abstract electronic word count: 275 words

#### Graphical abstract



#### Carvedilol is associated with improved survival in decompensated liver cirrhosis

#### Lay summary:

The safety of carvedilol and other non-selective beta blocker in liver cirrhosis with ascites is still debated. In this study, we have shown that carvedilol therapy in patients with liver cirrhosis and ascites was associated with reduced risk of mortality, particularly in those with mild ascites. We concluded that low dose, chronic treatment with carvedilol in patients with liver cirrhosis and ascites is not detrimental.

#### Abbreviations:

NSBB, non-selective beta-blocker; ALD, Alcoholic liver disease; NAFLD, Non-alcoholic fatty liver disease; HCC, Hepatocellular carcinoma; SBP, Spontaneous bacterial peritonitis; PSM, Propensity score matching; PT, Prothrombin time; INR, International Normalised Ratio; UKELD, United Kingdom Model for End Stage Liver Disease; MELD, Model for End Stage Liver Disease.

#### INTRODUCTION

Liver cirrhosis remains a leading cause of death worldwide. In the United Kingdom alone mortality from liver cirrhosis has risen dramatically in the last decade tandem with the rise in alcohol consumption and the epidemic of obesity [1]. Portal hypertension underlies many of its fatal complications. For this reason, non-selective beta-blocker (NSBB) was proposed to be beneficial in the management of patients with varices. Its unselective beta-blockade reduces cardiac output and splanchnic blood flow while the unopposed effect of alpha 1 receptors leads to splanchnic vasoconstriction thus reducing portal pressure and its attendant complications [2, 3]. Indeed, NSBB therapy has been associated with a 40% reduction in the risk of variceal bleeding and in combination with endoscopic ligation, the risk of rebleeding was lower than that seen with either therapy alone [4].

Lowering of portal pressure with NSBB therapy has also been shown to reduce the development of ascites, refractory ascites and hepatorenal syndrome [5]. Furthermore, the effect of NSBB on intestinal permeability, bacterial translocation and inflammatory response has been proposed to mitigate the risk of developing spontaneous bacterial peritonitis [6, 7]. Mookerjee *et al* have demonstrated an association between short-term survival with NSBB therapy and a significant reduction in inflammatory markers [7]. Its benefit on survival was further supported by the finding of higher in-hospital mortality among those with decompensated cirrhosis without NSBB compared with those who received NSBB irrespective of the ascites severity [8].

Carvedilol, an NSBB with intrinsic anti-alpha 1 receptor activity, has been associated with a greater reduction in portal pressure than the traditional beta blocker, propranolol [9]. In a meta-analysis, the proportion of patients achieving target haemodynamic response was higher with carvedilol than propranolol [9]. Additionally, the use of carvedilol has been shown to reduce portal pressure

in those who initially failed to respond to propranolol. Its action on alpha 1 receptor reduces portocollateral resistance, and its effects on hepatic stellate cells diminished intrahepatic resistance [10]. In a rodent model of liver injury, carvedilol has been reported to mitigate the development of fibrosis by modulating inflammatory cytokines and enhancing antioxidant response [11].

However, the safety of carvedilol and other NSBB in cirrhosis with ascites is still debated. Sersté *et al* reported poor survival among patients with refractory ascites on NSBB therapy [12]. Additionally, NSBB therapy has been associated with increased risk of paracentesis induced circulatory dysfunction [12, 13]. A potential mechanism underlying these observations is that NSBB can reduce cardiac output and perfusion, especially renal blood flow. In a milieu characterised by hemodynamic derangement associated with cirrhosis, such effect can potentially compromise organ perfusion leading to a multitude of complications including hepatorenal syndrome [14]. This led to the 'therapeutic window hypothesis' which posits that the development of refractory ascites marks the point at which beta-blockade has a detrimental effect on cardiac function and mean arterial pressure [15].

In clinical practice, this debate has led to uncertainty surrounding the role of NSBB and carvedilol in decompensated cirrhosis with ascites [16]. From a practical standpoint, the main question is whether carvedilol should be continued once ascites developed and progressed. We have previously shown that carvedilol at a low dose of 12.5 mg for four weeks is well tolerated and significantly reduced portal pressure without compromising systemic haemodynamics [3, 17]. In the present study, we sought to examine whether long-term use of carvedilol (therapy of more than four weeks) in the presence of ascites is a risk factor for mortality.

#### PATIENTS AND METHODS

#### Study population

This retrospective analysis included consecutive patients with cirrhosis and ascites who were admitted to the Liver Unit at The Royal Infirmary of Edinburgh between the  $1^{st}$  of January 2009 to the  $31^{st}$  of August 2012. The Royal Infirmary of Edinburgh serves as a tertiary referral centre for Lothian as well as a local hospital for Edinburgh city (population ~800,000). In our unit, carvedilol is the NSBB of choice for primary prophylaxis for oesophageal variceal haemorrhage (grade II or above on endoscopy) [18-21] and secondary prophylaxis if endoscopic band ligation is not tolerated or preferred by patient. Carvedilol is commenced at a low dose of 6.25 mg once daily and if tolerated, it is titrated to a target dose of 12.5 mg daily.

Patients were identified from a prospectively collected database. Date of the first radiologically confirmed ascites was taken as the time of entry into the study. The use of carvedilol was ascertained by history and medicine reconciliation obtained from electronic patient record (TrakCare, InterSystems Corp, MA, USA). Exclusion criteria were: non-liver causes of ascites, the use of selective beta-blocker (e.g. atenolol, bisoprolol) or other non-selective beta-blocker (e.g. propranolol), carvedilol therapy of less than four weeks and those who underwent liver transplantation (Figure 1). If carvedilol was discontinued during the study period, they remained in the carvedilol group as the chronic duration of the carvedilol treatment might have an impact on the long-term outcome.

#### Outcomes and follow-up

Primary outcomes measured were: all-cause mortality and liver-related mortality. All cases were followed up from the date of index presentation until the date of death or the 31st December 2015 whichever was earlier.

#### Data extraction

Demographic and imaging data were obtained from electronic integrated clinical records system (TrakCare, InterSystems Corp, MA, USA). The Information Services Division (ISD), NHS Scotland provided data linkage to the national dataset on mortality (National Records Scotland). The following variables were recorded: age, gender, aetiology of liver disease, previous variceal bleed, severity of ascites, laboratory data (serum creatinine, sodium, bilirubin, albumin, prothrombin time and international normalised ratio (INR)) and the use of antibiotics for primary or secondary spontaneous bacterial peritonitis (SBP) prophylaxis [20, 22]. Laboratory variables recorded were those obtained at the time of entry to the study. Severity of ascites was classified as mild, moderate or severe as documented in the radiology report.

Severity of liver disease was assessed using the UK model for End-stage Liver Disease (UKELD), which predicts annual mortality [23]. Given the retrospective nature of this analysis, Child-Turcotte-Pugh score was not included as the presence and the degree of hepatic encephalopathy was not reliably recorded. Model for End-Stage Liver Disease (MELD) was also calculated and compared between the groups, however, as carvedilol use of less than four weeks was excluded, the impact is somewhat biased.

#### Ethical approval

This study was considered as a retrospective review by the local committee in our institution hence no formal ethical approval from the NHS Research Ethics Committee was necessary. However, this study received approval from the Caldicott Guardian of NHS Lothian and the Caldicott Guardian of ISD for NHS Scotland.

#### Statistical analysis

Statistical analyses were performed using IBM SPSS Statistics for Macintosh, Version 21.0. (IBM Corp., Armonk, NY, USA), except for propensity score matching (PSM) analyses, which were performed on R, version 3.3.2 (The R Foundation for Statistical Computing, Vienna, Austria). The approach of Tukey's boxplot was followed to detect outliers with values outside the whiskers (2.2 times the interquartile range) being considered as outliers [24]. Outliers were checked and corrected if needed or retained where appropriate. Data were presented as mean (SD) for continuous variables and as frequencies and percentages for categorical variables. Kaplan-Meier survival analysis was used to calculate the cumulative incidence of primary outcomes. Cox regression models were fitted to adjust for effects of baseline characteristics and to determine their significance as a predictor for each outcome. Covariates with  $p \le 0.200$  in univariate regression analysis were included simultaneously in the multivariate models. The risk of the outcomes was presented as hazard ratio (95% confidence intervals). An unadjusted hazard ratio (HR) indicates an association between mortality and a single predictor. An adjusted HR indicates relationship between mortality and a predictor after adjusting for other independent variables. The proportional hazard assumptions were determined using log minus log plots for each variable. A p-value of less than 0.05 was deemed significant. To minimise the probability of selection bias, statistical analyses were performed in a cohort of carvedilol and non-carvedilol matched using the propensity score matching (PSM). The propensity score matching is a subject's probability of treatment, conditional on observed baseline covariates [25]. PSM is a recognised method of controlling for selection bias [26]. A PSM for carvedilol use versus non-carvedilol use was generated by multiple logistic regression. This model included all variables with clinical relevance to mortality (age, gender, aetiology, previous variceal bleed, severity of ascites, SBP antibiotic prophylaxis, serum sodium, serum creatinine, serum albumin, total bilirubin, International Normalised Ratio, and UK End Stage Liver Disease). The nearest available matching (1:1) on the estimated PSM method was used to construct the control group. Balance was achieved after

matching between the carvedilol (n=132) and non-carvedilol (n=132) groups on the selected confounders (Supplementary Data Table 1). The propensity score matching was performed using the MatchIt package in R [27, 28].

#### RESULTS

#### Clinical characteristics

Mean age for the entire cohort was 59.5 (SD 12.3) years and 67% were men. A total of 132 patients were on carvedilol with a median dose of 12.5 mg (range 6.25 -12.5 mg) per day. The discontinuation rate for carvedilol was 9% (12/132) in the first year and 80.3% (106/132) continued carvedilol therapy until the end of follow-up period. Mean UKELD and MELD score were 56 (SD 6.2) and 13.9 (SD 6.1) respectively. The prevalence of mild, moderate and severe ascites was 43.3%, 32.2% and 23.5% respectively. Baseline characteristics of the study population following the PSM are outlined in Table 1. The two groups were comparable with regards to aetiology of liver disease (p=0.91), previous variceal bleed (p=0.79), severity of ascites (p=0.93), antibiotic for SBP prophylaxis (p=0.41), serum sodium (p=0.41), serum creatinine (p=0.45), serum albumin (p=0.51), serum total bilirubin (p=0.71), INR (p=0.47), UKELD score (p=0.33) and MELD score (p=0.24).

#### Survival

The 264 patients were followed up for a median of 2.2 (IQR 0.8-4.1) years, during which there were 191 (72.3%) deaths. Among those on carvedilol, the estimated six-month and two-year survival was 95% and 61% respectively (Table 2). For non-carvedilol group, six-month survival was 89%, which declined to 47% at two-years. Similarly, survival at the end of the follow-up period was 24% for patients on carvedilol compared with only 2% among those who did not receive carvedilol (p=0.001). Median survival time was 1230 [CI 1006.9, 1453.1] days and 713 [CI 484.9, 941.0] days for carvedilol and non-carvedilol group respectively. After the first year, the unadjusted survival advantage associated with carvedilol was between 10 to 15% per year (Table 2).

As shown in Kaplan-Meier curves (Figure 2), the unadjusted hazard for carvedilol group was significantly lower than that seen in non-carvedilol group (Log Rank p=0.001). The use of carvedilol resulted in an unadjusted hazard ratio of 0.61 [CI 0.46, 0.81], when compared with those without carvedilol. The survival difference remained significant after adjusting for age, gender, previous variceal bleed, severity of ascites, aetiology of cirrhosis, SBP prophylaxis, serum albumin and UKELD with hazard ratio of 0.59 [CI 0.44, 0.80] suggesting 41% reduction of mortality risk associated with its use.

#### Ascites severity modifies the effect of carvedilol

Among those with mild ascites, the use of carvedilol resulted in an unadjusted hazard ratio of 0.51 [CI 0.33, 0.79]. The relationship remained significant even after adjusting for age, gender, aetiology of cirrhosis, previous variceal bleed, SBP prophylaxis, serum albumin and UKELD with 43% reduction in risk of death with carvedilol (HR 0.47 [CI 0.29, 0.77], p=0.003) when compared with those not on carvedilol. In moderate ascites, the trend of decreased risk of death with carvedilol was not statistically significant (p=0.08) with adjusted HR 0.61 [0.36, 1.06]. Similarly, we did not observe significant difference in the risk of death between carvedilol and non-carvedilol group among those with severe ascites (HR 0.77 [CI 0.42, 1.43], p=0.41) (Table 3).

#### Predictors of mortality

Table 4 demonstrates the findings from univariate and multivariate analyses. In univariate analysis, the following variables; carvedilol use, age, NAFLD as aetiology of liver cirrhosis, SBP prophylaxis were significantly associated with death. However, the effect of age was rendered insignificant after adjusting for other variables in a multivariate analysis (Table 4). In contrast, the association between nonalcoholic fatty liver disease and death remained significant with a two-fold increase in mortality when compared with alcoholic liver disease (HR 2.25 [CI 1.32, 3.84]). Additionally, antibiotic use for SBP prophylaxis was an independent predictor of mortality (p=0.007).

#### *Causes of death*

There were 191 deaths in the entire cohort (72.3%; 191/264). Of these 90.6% (173/191) were liver-related deaths. The leading cause of liver-related death was alcoholic liver disease (ALD) (53.9%; 103/191) followed by variceal bleed (14.1%; 27/191), hepatocellular carcinoma (HCC) (10.5%; 20/191) and Non-alcoholic fatty liver disease (NAFLD) (9.4%; 18/191).

ALD related deaths for carvedilol and non-carvedilol group were 48.2% (40/83) and 41.7% (43/108) respectively (p=0.66). The proportion of deaths relating to variceal bleed (16.9%; 14/83) for carvedilol and (12%; 13/108) for non-carvedilol group (p=0.40). Of the 14 patients who died of variceal haemorrhage, eight were on 12.5 mg of carvedilol and the remaining were on 6.25 mg of carvedilol. Similarly, there was no difference in HCC, NAFLD, viral and sepsis related deaths between the two groups (Table 5).

The leading cause of non-liver related deaths was cardiovascular disorders (18.8%; 36/191). Cardiovascular related deaths occurred in 16.9% (14/83) of patients on carvedilol and in 20.4% (22/108) of those in non-carvedilol group (p=0.58). We did not observe significant difference in cardiovascular-related death between patients with NAFLD on carvedilol and those with NAFLD without carvedilol. Similarly, there was no difference in deaths relating to extrahepatic malignancy, respiratory or renal disorders between the carvedilol and non-carvedilol groups (Table 5).

#### DISCUSSION

In this study, we examined for the first time whether long-term, low dose carvedilol therapy influences mortality in established cirrhosis with ascites. It also addressed an important clinical question-should carvedilol be continued in the presence of ascites. Our data highlight two important points. First, we have shown that long-term use of low dose carvedilol in cirrhosis with ascites was associated with reduced risk of mortality. Second, this association was modified by the severity of ascites; carvedilol therapy was associated with decreased risk of death in mild ascites but did not affect the outcome of those with moderate or severe ascites. Importantly, even in the presence of severe ascites, carvedilol therapy was not associated with excess mortality.

Our findings are in agreement with previous studies which largely focused on the role of propranolol in cirrhosis. The use of NSBB in cirrhosis with ascites has been associated with improved survival in various population ranging from hospital admission [8] to those on transplant waiting list [29]. Our study exclusively examined the impact of carvedilol in the presence of ascites. Carvedilol is the NSBB of choice in the preventative strategy against variceal bleed in our centre. Its safety in decompensated cirrhosis with ascites is still debated, for instance, its use in two of the studies examined in a recent meta-analysis was associated with increased mortality [30].

It was postulated that the vasodilatory effect from the concomitant alpha-blockade can potentially exacerbate arterial hypotension in the already precarious haemodynamic state. The effects of carvedilol on systemic haemodynamics are dose dependent [3, 17] and we have previously shown that carvedilol at 12.5 mg for four weeks led to a marked reduction in portal pressure without significantly altering the mean arterial pressure [3]. In the present study, carvedilol therapy at a median dose of 12.5 mg per day for a median duration of 49 months was associated with reduced risk of death. The estimated survival throughout the follow-up period was favourable with

carvedilol. Indeed, the survival at one-year was 78% for carvedilol and 65% for non-carvedilol despite the similarities in the baseline UKELD score. These groups were also similar with regards to liver disease severity, aetiology of liver disease and the degree of ascites.

The disparities between our findings and that of Njei *et al* may also have been attributed to the differences in population studied [30]. Specifically, our study population comprised all grades of ascites rather than solely focused on refractory ascites. Our data suggest that the severity of ascites modifies the effect of carvedilol. The use of carvedilol in mild ascites was associated with 53% reduced risk of long-term mortality. Although the risk of death in moderate and severe ascites was statistically similar between carvedilol and non-carvedilol group, the trend was that the impact of carvedilol decreased with increasing severity of ascites. This is consistent with the 'therapeutic window hypothesis' which stipulates that NSBB therapy may be effective only within a particular clinical window of advanced liver disease [15].

The underlying mechanisms for the protective effect of carvedilol in our study is unclear. It is plausible that similar to the traditional NSBB, the benefits incurred hinge on its portal hypotensive effect. Despite the similarities in variceal bleed related mortality between the two groups, the long-term effects of its portal pressure lowering on ascites related complications, bacterial translocation, anti-inflammatory and anti-fibrotic cannot be excluded [7, 11]. The cause of death documented reflects the progression of the underlying liver disease compounded by concurrent comorbidities, in particular, ischaemic heart disease.

Some limitations must be acknowledged. First, the retrospective nature of this study precludes the analysis of spontaneous bacterial peritonitis, evidence of autonomic dysfunction, interaction of concomitant drugs, blood pressure measurements, hepatic encephalopathy, ongoing alcohol consumption and compliance with carvedilol. Second, degree of ascites severity was based on radiology reports, which was subject to operator bias. Additionally, we did not have a reliable information on the clinical course of ascites following the treatment with carvedilol. Similarly, we do not have reliable data regarding the safety of carvedilol in patients with refractory ascites. Information regarding concomitant use of diuretics and paracentesis were also not available.

However, our cohort represents real-world insight of patients with established cirrhosis and ascites. We used propensity score matching to minimise any potential bias. Our data have important implications for clinical management of patients with established cirrhosis and ascites. Long-term treatment with carvedilol at a low dose, even in the presence of severe ascites, is not associated with increased mortality. Therefore, our data support continuing carvedilol therapy in patients with ascites unless there are features to suggest significant haemodynamic perturbation compromising organ perfusion. As such, prospective randomised controlled trial is required to evaluate the efficacy of carvedilol in well characterised spectrum of cirrhosis and ascites. Further studies on the effects of carvedilol on the underlying disease progression and systemic inflammatory response modulation might unravel its effect on long-term survival.

In conclusion, we have demonstrated that low dose, chronic treatment with carvedilol in patients with decompensated cirrhosis and ascites is not detrimental. Furthermore, we have shown that carvedilol therapy in this setting was associated with reduced mortality, particularly in those with mild ascites. Our data support the need of prospective studies to determine its impact in severe and refractory ascites.

### Acknowledgment

We are immensely grateful to Dr JJ Kerssens of the Farr Institute, Scotland for his statistical advice. Authors are also grateful to Catherine Bisset for her assistance in maintaining the Liver Unit database.

#### REFERENCES

[1] Williams R, Horton R. Liver disease in the UK: a Lancet Commission. Lancet 2013;382:1537-1538.

[2] Frishman WH. Carvedilol. N Engl J Med 1998;339:1759-1765.

[3] Tripathi D, Therapondos G, Lui HF, Stanley AJ, Hayes PC. Haemodynamic effects of acute and chronic administration of low-dose carvedilol, a vasodilating beta-blocker, in patients with cirrhosis and portal hypertension. Aliment Pharmacol Ther 2002;16:373-380.

[4] Gonzalez R, Zamora J, Gomez-Camarero J, Molinero LM, Banares R, Albillos A. Meta-analysis: Combination endoscopic and drug therapy to prevent variceal rebleeding in cirrhosis. Ann Intern Med 2008;149:109-122.

[5] Hernandez-Gea V, Aracil C, Colomo A, Garupera I, Poca M, Torras X, et al. Development of ascites in compensated cirrhosis with severe portal hypertension treated with beta-blockers. Am J Gastroenterol 2012;107:418-427.

[6] Senzolo M, Cholongitas E, Burra P, Leandro G, Thalheimer U, Patch D, et al. beta-Blockers protect against spontaneous bacterial peritonitis in cirrhotic patients: a meta-analysis. Liver Int 2009;29:1189-1193.

[7] Mookerjee RP, Pavesi M, Thomsen KL, Mehta G, Macnaughtan J, Bendtsen F, et al. Treatment with non-selective beta blockers is associated with reduced severity of systemic inflammation and improved survival of patients with acute-on-chronic liver failure. J Hepatol 2016;64:574-582.

[8] Aday AW, Mayo MJ, Elliott A, Rockey DC. The Beneficial Effect of Beta-Blockers in Patients With Cirrhosis, Portal Hypertension and Ascites. Am J Med Sci 2016;351:169-176.

[9] Li T, Ke W, Sun P, Chen X, Belgaumkar A, Huang Y, et al. Carvedilol for portal hypertension in cirrhosis: systematic review with meta-analysis. BMJ Open 2016;6:e010902.

[10] Tripathi D, Stanley AJ, Hayes PC, Patch D, Millson C, Mehrzad H, et al. U.K. guidelines on the management of variceal haemorrhage in cirrhotic patients. Gut 2015;64:1680-1704.

[11] Araujo Junior RF, Garcia VB, Leitao RF, Brito GA, Miguel Ede C, Guedes PM, et al. Carvedilol Improves Inflammatory Response, Oxidative Stress and Fibrosis in the Alcohol-Induced Liver Injury in Rats by Regulating Kuppfer Cells and Hepatic Stellate Cells. PLoS One 2016;11:e0148868.

[12] Serste T, Melot C, Francoz C, Durand F, Rautou PE, Valla D, et al. Deleterious effects of betablockers on survival in patients with cirrhosis and refractory ascites. Hepatology 2010;52:1017-1022.

[13] Serste T, Francoz C, Durand F, Rautou PE, Melot C, Valla D, et al. Beta-blockers cause paracentesis-induced circulatory dysfunction in patients with cirrhosis and refractory ascites: a cross-over study. J Hepatol 2011;55:794-799.

[14] Wong F, Salerno F. Beta-blockers in cirrhosis: friend and foe? Hepatology 2010;52:811-813.

[15] Krag A, Wiest R, Albillos A, Gluud LL. The window hypothesis: haemodynamic and non-haemodynamic effects of beta-blockers improve survival of patients with cirrhosis during a window in the disease. Gut 2012;61:967-969.

[16] Thorhauge KH, Lindvig KP, Laleman W, Angeli P, Singh SP, Krag A. Lack of consensus for usage of beta-blockers in end-stage liver disease. Gut 2016.

[17] Stanley AJ, Therapondos G, Helmy A, Hayes PC. Acute and chronic haemodynamic and renal effects of carvedilol in patients with cirrhosis. J Hepatol 1999;30:479-484.

[18] Tripathi D, Ferguson JW, Kochar N, Leithead JA, Therapondos G, McAvoy NC, et al. Randomized controlled trial of carvedilol versus variceal band ligation for the prevention of the first variceal bleed. Hepatology 2009;50:825-833.

[19] Garcia-Tsao G, Bosch J. Management of varices and variceal hemorrhage in cirrhosis. N Engl J Med 2010;362:823-832.

[20] de Franchis R, Baveno VF. Revising consensus in portal hypertension: report of the Baveno V consensus workshop on methodology of diagnosis and therapy in portal hypertension. J Hepatol 2010;53:762-768.

[21] de Franchis R, Baveno VIF. Expanding consensus in portal hypertension: Report of the Baveno VI Consensus Workshop: Stratifying risk and individualizing care for portal hypertension. J Hepatol 2015;63:743-752.

[22] Fernandez J, Navasa M, Planas R, Montoliu S, Monfort D, Soriano G, et al. Primary prophylaxis of spontaneous bacterial peritonitis delays hepatorenal syndrome and improves survival in cirrhosis. Gastroenterology 2007;133:818-824.

[23] Neuberger J, Gimson A, Davies M, Akyol M, O'Grady J, Burroughs A, et al. Selection of patients for liver transplantation and allocation of donated livers in the UK. Gut 2008;57:252-257.

[24] Hoaglin DC, Iglewicz, B. Fine tuning some resistant rules for outlier labeling. Journal of American Statistical Association 1987;82:1147-1149.

[25] Austin PC. Balance diagnostics for comparing the distribution of baseline covariates between treatment groups in propensity-score matched samples. Stat Med 2009;28:3083-3107.

[26] D'Agostino RB, Jr. Propensity score methods for bias reduction in the comparison of a treatment to a non-randomized control group. Stat Med 1998;17:2265-2281.

[27] Randolph JJ FK, Manuel AK, Balloun JL. A step-by-step guide to propensity score matching in R. Practical Assessment, Research & Evaluation 2014;19:2.

[28] Ho DE IK, King G, Stuart EA. MatchIt: Nonparametric preprocessing for parametric causal inference. Journal of Statistical Software 2011;8:1-28.

[29] Leithead JA, Rajoriya N, Tehami N, Hodson J, Gunson BK, Tripathi D, et al. Non-selective beta-blockers are associated with improved survival in patients with ascites listed for liver transplantation. Gut 2015;64:1111-1119.

[30] Njei B, McCarty TR, Garcia-Tsao G. Beta-blockers in patients with cirrhosis and ascites: type of beta-blocker matters. Gut 2016.

	Pre-match			Post-match		
	Carvedilol	Non-		Carvedilol	Non-	
Clinical characteristics	( <i>n</i> =132)	group ( <i>n</i> =193)	<i>p</i> -value	( <i>n</i> =132)	group ( <i>n</i> =132)	<i>p</i> -value
Age (years)	59.1 (12.7)	59.9 (12.0)	0.55	59.1 (12.7)	59.8 (11.9)	0.62
Men	87 (65.9)	116 (60.1)	0.29	87 (65.9)	89 (67.4)	0.89
Aetiology			0.31			0.91
ALD NAFLD Viral Others HCC	94 (71.2) 15 (11.4) 5 (3.8) 12 (9.1) 6 (4.5)	157 (81.3) 15 (7.8) 4 (2.1) 12 (6.2) 5 (2.6)		94 (71.2) 15 (11.4) 5 (3.8) 12 (9.1) 6 (4.5)	101 (76.5) 12 (9.1) 4 (3.0) 10 (7.6) 5 (3.8)	
Previous variceal bleed	42 (31.8)	66 (34.2)	0.72	42 (31.8)	45 (34.1)	0.79
Severity of ascites Mild Moderate Severe	60 (45.5) 42 (31.8) 30 (22.7)	67 (34.7) 69 (35.8) 57 (29.5)	0.13	60 (45.5) 42 (31.8) 30 (22.7)	57 (43.2) 43 (32.6) 32 (24.2)	0.93
Antibiotic for SBP	40 (30.3)	34 (17.6)	0.01*	40 (30.3)	33 (25.0)	0.41
Serum sodium (mmol/L)	134.8 (5.1)	134.6 (5.2)	0.81	134.8 (5.1)	134.3 (5.5)	0.41
Serum creatinine	71.8 (31.2)	77.5 (51.1)	0.26	71.3 (26.2)	74.3 (37.8)	0.45
Serum albumin (g/L)	28.6 (7.8)	28.0 (6.9)	0.48	28.1 (5.6)	27.6 (6.9)	0.51
Total bilirubin (mmol/L)	87.8 (107.1)	88.4 (106.2)	0.96	88.3 (108.3)	93.2 (104.3)	0.71
INR	1.5 (0.5)	1.5 (0.5)	0.55	1.5 (0.4)	1.6 (0.5)	0.47
UKELD	55.7 (6.1)	55.9 (6.1)	0.76	55.7 (6.2)	56.5 (6.3)	0.33
MELD	12 (9)	13 (9)	0.26	13 (6)	14 (6)	0.24

	Estimated survival in percentage		
Time interval	Carvedilol	Non-carvedilol	
30-day	98	95	
6 month	95	89	
1-year	78	65	
2-year	61	47	
3-year	55	40	
4-year	39	27	
5-year	33	20	

Severity of ascites	Carvedilol	Non-carvedilol	<i>p</i> -value
Mild			-
Mortality $(n, \%)$	35 (58.3)	50 (87.7)	$0.000^{*}$
Unadjusted HR	0.51 (0.33 - 0.79)	Ref	$0.003^{*}$
Adjusted HR	0.47(0.29 - 0.77)	Ref	$0.003^{*}$
Moderate			
Mortality $(n, \%)$	26 (61.9)	34 (79.1)	0.09
Unadjusted HR	0.59(0.35 - 0.99)	Ref	$0.05^{*}$
Adjusted HR	0.61 (0.36 – 1.06)	Ref	0.08
Severe			
Mortality $(n, \%)$	22 (73.3)	24 (75.0)	1.00
Unadjusted HR	0.90(0.50 - 1.61)	Ref	0.73
Adjusted HR	0.77 (0.42 – 1.43)	Ref	0.41

	Univariate analysis		Multivariate analysis	
	Hazard ratio		Adjusted hazard ratio	
Variables	(95% confidence interval)	<i>p</i> -value	(95% confidence interval)	Adjusted <i>p</i> -value
Carvedilol	0.61 (0.46 – 0.81)	0.001*	0.59 (0.44 – 0.80)	0.001*
Age	1.01 (1.00-1.02)	0.04*	1.00 (0.99 – 1.02)	0.34
Men	1.09 (0.81 – 1.48)	0.55	1.06 (0.77 – 1.47)	0.69
Aetiology				
ALD	Ref		Ref	
NAFLD	1.82 (1.18 – 2.81)	0.007 *	2.25 (1.32 – 3.84)	0.003*
НСС	0.44 (0.14 – 1.39)	0.16	0.51 (0.16 – 1.64)	0.26
Viral	0.91 (0.52 – 1.59)	0.75	1.14 (0.63 – 2.07)	0.65
Others	1.85 (0.90 - 3.78)	0.09	1.87 (0.89 – 3.94)	0.09
Previous variceal bleed	0.80 (0.52 - 1.09)	0.16	0.77 (0.56 - 1.06)	0.11
Severity of ascites				
Mild	Ref		Ref	
Moderate	1.06 (0.76 – 1.48)	0.71	0.93 (0.65 – 1.33)	0.66
Severe	1.20 (0.84 – 1.72)	0.31	1.26 (0.87 – 1.84)	0.22
SBP prophylaxis	0.64 (0.45- 0.89)	0.009*	0.62 (0.43 – 0.88)	0.007*
Serum sodium	1.01 (0.98 – 1.04)	0.37		
Serum creatinine	1.00 (0.99 – 1.00)	0.88		
Serum albumin	0.98 (0.95 – 1.00)	0.063	0.98 (0.95 - 1.00)	0.09
Total bilirubin	1.00 (0.99 – 1.00)	0.67		
INR	1.09 (0.81 – 1.45)	0.57		
UKELD	1.00 (0.98 - 1.02)	0.93	0.46 (1.01 – 0.98)	0.46
MELD	1.00 (0.97 – 1.32)	0.44		

Cause of death	Carvedilol ( <i>n</i> =83)	Non-carvedilol ( <i>n</i> =108)	<i>p</i> value
	n (%)	n (%)	
Liver related			
ALD	43 (51.8)	60 (55.6)	0.66
Variceal bleeding	14 (16.9)	13 (12.0)	0.40
Sepsis	12 (14.5)	20 (18.5)	0.56
NĂFLD	7 (8.4)	11 (10.2)	0.80
HCC	11 (13.3)	9 (8.3)	0.34
Viral	9 (10.8)	9 (8.3)	0.62
Others	29 (34.9)	28 (25.9)	0.20
Non-liver related			
Vascular event	14 (16.9)	22 (20.4)	0.58
Respiratory	12 (14.5)	19 (17.6)	0.69
Renal	5 (6.0)	6 (5.6)	1.00
Extra-hepatic malignancy	6 (7.2)	8 (7.4)	1.00









#### Figure and Table Legends

**Table 1: Baseline characteristics of study population.** Data presented as mean and standard deviation (SD) or frequency (*n*) and percentage where appropriate. \*p<0.05 is significant. ALD, Alcoholic liver disease; NAFLD, Non-alcoholic fatty liver disease; Viral, Viral hepatitis; HCC, Hepatocellular carcinoma; SBP, Spontaneous bacterial peritonitis; PT, Prothrombin time; INR, International Normalised Ratio; UKELD, United Kingdom Model for End Stage Liver Disease.

#### Table 2: Estimated survival over time.

Table 3: Risk of mortality stratified by the severity of ascites. Adjusted hazard ratio is obtained from Cox-regression analysis adjusted to the following variables (carvedilol, age, gender, aetiology, severity of ascites, previous variceal bleed, SBP prophylaxis, serum albumin and UKELD); Data presented as frequency (*n*) and percentage of each study group or hazard ratio (HR) and 95% confidence interval. \**p* <0.05 is significant; Ref, reference group. SBP, Spontaneous bacterial peritonitis; UKELD, United Kingdom Model for End Stage Liver Disease.

Table 4: Independent predictors of mortality across study population. Unadjusted and adjusted hazard ratios are obtained from univariate and multivariate analysis respectively. The following variables were considered in multivariate analysis (carvedilol, age, gender, aetiology, severity of ascites, previous variceal bleed, SBP prophylaxis, serum albumin and UKELD); Data presented as hazard ratio (HR) and 95% confidence interval. \**p* <0.05 is significant; Ref, reference group. SBP, Spontaneous bacterial peritonitis; UKELD, United Kingdom Model for End Stage Liver Disease.

**Table 5: Cause of death.** Data presented as frequency (*n*) and percentage of each study group. ALD, Alcoholic liver disease; NAFLD, Non-alcoholic fatty liver disease; Viral, Viral hepatitis; HCC, Hepatocellular carcinoma. \*p < 0.05 is significant.

Figure 1: Study flow.

Figure 2: Kaplan-Meier curves for carvedilol and non-carvedilol groups. Carvedilol therapy was associated with lower hazard compared with non-carvedilol group over the follow-up period. Log Rank (Mantel-Cox) p value of 0.001; \*p<0.05 is significant.